Author’s response to reviews

Title: The Active for Life Year 5 (AFLY5) school based cluster randomised controlled trial: study protocol for a randomized controlled trial.

Authors:

Debbie A Lawlor (d.a.lawlor@bristol.ac.uk)
Russell Jago (russ.jago@bristol.ac.uk)
Sian M Noble (s.m.noble@bristol.ac.uk)
Catherine R Chittleborough (catherine.chittleborough@bristol.ac.uk)
Rona Campbell (rona.campbell@bristol.ac.uk)
Julie Mytton (julie.mytton@bristol.nhs.uk)
Laura Howe (laura.howe@bristol.ac.uk)
Tim Peters (tim.peters@bristol.ac.uk)
Ruth R Kipping (ruth.kipping@bristol.ac.uk)

Version: 4 Date: 17 July 2011

Author’s response to reviews:

Dear Editor,

In response to the email from the independent editorial production team (14th July) the following revisions have been made to the manuscript:

1. Reference 18 has been updated in the text and reference list

2. The figure legend has been added to the manuscript as requested and the figure uploaded as a separate document

3. The references have been fully checked and corrected so that all references now have corresponding numbers in the text (in the correct order) and all have full author lists (no et al.). We have also updated one reference that was ‘in press’ and has now been published.

4. We have checked the manuscript in full and corrected all typographical errors and also updated the current status of the study section.